Author's response to reviews

Title: Constitution-specific features of perspiration and skin visco-elasticity in SCM

Authors:

Young-Min Kim (irobo77@kiom.re.kr)
Boncho Ku (secondmoon@kiom.re.kr)
Chang Jin Jung (traviswas@kiom.re.kr)
Jaeuk U. Kim (jaeukkim@kiom.re.kr)
Young Ju Jeon (jyi92@kiom.re.kr)
Keun Ho Kim (rkim70@kiom.re.kr)
Jong Yeol Kim (ssmed@kiom.re.kr)

Version: 6 Date: 19 December 2013

Author's response to reviews: see over
We appreciate editorial comments. According to their comments, we revised our paper as follows.

(Editorial Comments) Please confirm whether informed consent to take part was obtained from the participants in your study. Please include a statement to this effect in your manuscript.

(Answer) We confirmed the informed consent to take part of all participants and included the following statement in ‘Methods’ section.

 Approval of the clinical trials was obtained from the institutional review board at the oriental hospital of Daejeon university, Korea (IRB No, M2012-01) and written informed consent to take part was obtained from all participants in this study. (Page 6, 9th line from top)